

Citation: Frenoy A, Bonhoeffer S (2018) Death and population dynamics affect mutation rate estimates and evolvability under stress in bacteria. PLoS Biol 16(5): e2005056. <u>https://doi.org/10.1371/journal.pbio.2005056</u>

Academic Editor: Arjan de Visser, Wageningen Universiteit en Researchcentrum, Netherlands

Received: December 6, 2017

Accepted: April 12, 2018

Published: May 11, 2018

Copyright: © 2018 Frenoy, Bonhoeffer. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The full dataset (including raw data, simulation programs, data analysis programs, and processed data; permitting a full reconstruction of the figures from the raw data) is available on zenodo (doi: <u>10.5281/zenodo.</u> <u>1211765</u>).

Funding: European Research Council <u>https://</u> <u>cordis.europa.eu/project/rcn/99462_en.html</u> (grant number 268540). Received by SB. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. Swiss National Science Foundation RESEARCH ARTICLE

Death and population dynamics affect mutation rate estimates and evolvability under stress in bacteria

Antoine Frenoy*, Sebastian Bonhoeffer

Institute for Integrative Biology, ETH Zürich, Zurich, Switzerland

* antoine.frenoy@pasteur.fr

Abstract

The stress-induced mutagenesis hypothesis postulates that in response to stress, bacteria increase their genome-wide mutation rate, in turn increasing the chances that a descendant is able to better withstand the stress. This has implications for antibiotic treatment: exposure to subinhibitory doses of antibiotics has been reported to increase bacterial mutation rates and thus probably the rate at which resistance mutations appear and lead to treatment failure. More generally, the hypothesis posits that stress increases evolvability (the ability of a population to generate adaptive genetic diversity) and thus accelerates evolution. Measuring mutation rates under stress, however, is problematic, because existing methods assume there is no death. Yet subinhibitory stress levels may induce a substantial death rate. Death events need to be compensated by extra replication to reach a given population size, thus providing more opportunities to acquire mutations. We show that ignoring death leads to a systematic overestimation of mutation rates under stress. We developed a system based on plasmid segregation that allows us to measure death and division rates simultaneously in bacterial populations. Using this system, we found that a substantial death rate occurs at the tested subinhibitory concentrations previously reported to increase mutation rate. Taking this death rate into account lowers and sometimes removes the signal for stress-induced mutagenesis. Moreover, even when antibiotics increase mutation rate, we show that subinhibitory treatments do not increase genetic diversity and evolvability, again because of effects of the antibiotics on population dynamics. We conclude that antibiotic-induced mutagenesis is overestimated because of death and that understanding evolvability under stress requires accounting for the effects of stress on population dynamics as much as on mutation rate. Our goal here is dual: we show that population dynamics and, in particular, the numbers of cell divisions are crucial but neglected parameters in the evolvability of a population, and we provide experimental and computational tools and methods to study evolvability under stress, leading to a reassessment of the magnitude and significance of the stressinduced mutagenesis paradigm.

http://p3.snf.ch/project-155866 (grant number 155866). Received by SB. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

 Abbreviations: ATREYU, Approximate bayesian

 computing Tentative mutation Rate Estimator that

 You could Use; CFSE, Carboxyfluorescein

 succinimidyl ester; CFU, colony-forming unit;

 FALCOR, Fluctuation AnaLysis CalculatOR; IPTG,

 Isopropyl β-D-1-thiogalactopyranoside; MIC,

 minimal inhibitory concentration; pgf, probability

 generating function.

Author summary

The effect of environmental stress on bacterial mutagenesis has been a paradigm-shift discovery. Recent developments include evidence that various antibiotics increase mutation rates in bacteria when used at subinhibitory concentrations. It is therefore suggested that such treatments promote resistance evolution because they increase the generation of genetic variation on which natural selection can act. However, existing methods to compute mutation rate neglect the effect of stress on death and population dynamics. Developing new experimental and computational tools, we find that taking death into account significantly lowers the signal for stress-induced mutagenesis. Moreover, we show that treatments that increase mutation rate do not always lead to increased genetic diversity, which questions the standard paradigm of increased evolvability under stress.

Introduction

One of the most puzzling and controversial microbial evolution experiments of the 20th century may be the one performed by Cairns and colleagues [1,2] in which *lac*- cells are plated on lactose as the sole carbon source and therefore cannot grow. Revertants toward the *lac*+ genotype continuously appear after plating at a rate and timing seemingly incompatible with the Darwinian hypothesis of selection of preexisting mutants. In the *lac*- construct, the *lacZ* coding sequence is present but nonfunctional, because it is out of frame with the start codon. The *lac*+ revertants are thus frameshift mutants in which this coding sequence is back in frame with the start codon. Most of the controversy initially came from the question of whether these reversion mutations were Lamarckian, in the sense that they would arise at a higher rate when the cells would "sense" that these mutations would be beneficial [3]. However, many additional experiments quickly suggested that this phenomenon can be explained by more standard Darwinian mechanisms, in which genetic changes are not targeted but occur randomly and are then selected or not. While two seemingly conflicting molecular explanations—the stressinduced mutagenesis model and the gene amplification model—emerged, both are conceptually very similar.

In both explanations, mutations occur randomly and independently of their effect on fitness, but the specific conditions of carbon starvation increase the rate at which genetic diversity is generated at the relevant locus (*lacI-lacZ* sequence). In the stress-induced mutagenesis model [4], the genome-wide mutation rate is increased as an effect of the stress response triggered by starvation. In the gene amplification model [5,6], random duplications of the *lacIlacZ* system happen before plating on lactose and are then selected in presence of lactose because the frameshift mutation is leaky. A small amount of Beta-galactosidase is still synthesized, permitting cryptic growth due to rare expression errors, which compensate the frameshift. This residual expression becomes higher with more copies of the leaky system. As the copy number of the system increases, a reversion mutation in *lacI-lacZ* becomes more likely because of increased target number.

While it is still not clear whether stress-induced mutagenesis is the sole explanation of the phenomenon, the attempts to explain the data presented by Cairns and colleagues have led to a much better understanding of control over mutation rate in response to the environment. An increase in mutation rate under starvation has also been reported in other systems, such as nutrient-limited liquid cultures [7,8] and "aging" colonies on agar plates [9,10]. However, Wrande and colleagues [11] reported that the accumulation of mutants in aging colonies observed by Bjedov and colleagues [10] can be explained by growth under selection without

elevated mutagenesis, because the mutation used by the original authors to infer mutation rate is beneficial in these specific environmental conditions.

An emblematic molecular mechanism permitting this regulation of mutation rate is the SOS response (suggested in 1970 [12,13]), in which DNA damage is sensed by bacterial cells and leads to the up-regulation of many genes, permitting mutagenic repair and replication of damaged DNA. While the responsible enzymes were unknown at the time, it has indeed been found subsequently that the SOS response increases the dosage of polIV and polV [14,15]. These error-prone polymerases are able to replicate damaged DNA that the classical DNA polymerase polIII could not replicate, albeit at the price of a higher error rate [16,17]. This strategy, favoring "survival at the price of the mutation," is only one side of the story. There is a line of evidence suggesting that this higher error rate is not only an unavoidable trade-off with survival. It is also supposed to be a selected property to increase mutation rate under stressful conditions, increasing the chances that one of the descendants obtains a beneficial mutation that makes it able to better withstand the stress [18,19].

The evolution of traits that increase mutation rate under stress needs be considered in the context of second-order selection [20]. Second-order selection relies on the idea that natural selection does not only act on the individual's phenotype and instant fitness but also on its ability to generate fit descendants, leading to selection of properties such as evolvability and mutational robustness [21]. In parallel to the study of environmental control over the mutation rate, genetic determinants of mutation rate have also been studied. It has been shown and is widely accepted that alleles increasing mutation rate (for example, defective mismatch repair or DNA proofreading) can be selected when hitchhiking with the beneficial mutations they permit to generate [22,23]. On the other hand, the possibility of selection of mechanisms increasing mutation rate under stress but not constitutively has been subject to a more philosophical debate [24,25]. While modeling shows such selection is possible [19], it is hard to distinguish whether an observed increase in mutation rate under a specific stress is (i) an evolvability strategy; (ii) an unavoidable trade-off of selection for survival, such as replicating damaged DNA to avoid death at the price of making mutations; or (iii) a direct effect of the stress and not of the stress-response system [26].

This debate is important for a full understanding of the evolutionary relevance of the phenomenon but does not affect the medical implications concerning the risk of de novo evolution of resistance during antimicrobial treatment [27]. Here, we are interested in the general case of mutation rate in growing stressed populations, and we especially focus on antibiotic stress, although our findings may be valid for other biotic and abiotic stresses. It has been suggested that treatment with subinhibitory doses of antibiotics increases bacterial mutation rate, due to induction of various stress-response pathways [28–32]. Many molecular mechanisms underlying this stress response have been elucidated, including the SOS response [29] or the RpoS regulon [30]. Oxidative damage has also been suggested to play a role in antibioticinduced mutagenesis [28] and death [33–35]. Although still controversial [36], these findings link antibiotic stress to the older question of how bacteria deal with oxidative stress and how oxidative damage impacts mutation rates [37].

However, all the evidence for stress-induced mutagenesis relies on accurately measuring mutation rates of bacteria growing in stressful conditions, and comparing them to those of the same strains growing without stress. Computing such mutation rates under stress is harder than it may seem, because stress may change population dynamics and may thus invalidate the assumptions made by the mathematical models used to compute mutation rate. For example, in the case of subinhibitory concentrations of antibiotics in which net population growth is positive, death may nevertheless happen at a considerable rate. Death events, however, are not detected by standard microbiology methods and are not taken into account by the mathematical tools used to compute mutation rate [38–40].

Indeed, such tools take as inputs only the number of observed mutants at a chosen locus and the final population size, making the underlying assumption that there is no death and that population size is thus sufficient information to summarize growth dynamics. The final population size is used to infer the number of DNA divisions leading to the final observed population from a small initial inoculum. If there is death, more divisions are needed to reach this population size, thus giving more opportunities to acquire mutations. The mutation rate will then be overestimated, because the number of DNA replications will be underestimated.

In this work, we developed an experimental system to compute death rates in populations growing under stress and a computational method to compute mutation rates from fluctuation assays under stress using the computed death rates. We applied this framework to re-estimate mutation rates of *Escherichia coli* MG1655 growing under sub- minimal inhibitory concentration (MIC) doses of kanamycin (an aminoglycoside acting on protein synthesis [41]), norfloxacin (a fluoroquinolone acting on the DNA-gyrase complex and potentially leading to the creation of DNA breaks through the cell machinery [42]), and hydrogen peroxide (an oxidizing agent producing reactive oxygen species that directly affect DNA independently of the cell machinery [43,44]). All these antimicrobials have previously been reported to significantly elevate mutation rate [28,31]. We find the same pattern when computing mutation rate without taking death into account. However, for norfloxacin and kanamycin, the estimated increase of mutation rate due to treatment is strongly reduced after conservatively correcting for death. There remains no signal of stress-induced mutagenesis in the case of kanamycin. These findings confirm our suspicion that neglecting death leads to substantial overestimation of mutation rate under stress.

We also show that mutation rate estimation does not only present experimental and mathematical challenges; it is also not the most relevant measure of evolvability, meaning the capacity of a population to generate adaptive genetic diversity. Indeed, some of the studied subinhibitory treatments cause a significant drop in population size due to both bactericidal and bacteriostatic effects and thus lead to a smaller genetic diversity despite a higher mutation rate. Ironically, evolvability, approximated as the generation of genetic diversity, can be much more easily estimated from experimental data than mutation rate. In our experiments, antibiotics and hydrogen peroxide have very different effects on evolvability: both subinhibitory norfloxacin and kanamycin treatments significantly reduce it, while hydrogen peroxide treatment strongly increases it.

Results

Mutation rates are overestimated when neglecting death

Subinhibitory treatments are not necessarily sublethal, because minimal inhibitory concentration is defined at population scale. An antimicrobial treatment is subinhibitory if the population grows (i.e., colony-forming unit [CFU]/mL increases or, more crudely, culture tubes inoculated at low density are turbid after 24 h). However, the death rate can be high, as long as the division rate is higher. Such death events will not be visible to the observer if only population size (CFU/mL) is tracked over time (Fig 1). To reach a given observed final population size, the number of cell divisions has to be higher if there is death. This means that when computing mutation rate using the classical approach (described in the <u>Materials and methods</u>), the number of cell divisions will be underestimated. This is because it is implicitly assumed that there is no death and thus that the final population size is a good approximation for the number of cell divisions. The mutation rate, computed as the number of mutational events divided by the number of cell divisions, will then be systematically overestimated.

The above statement—that mutation rates are systematically overestimated when there is death—is the first intuition motivating our work. We explore this intuition more rigorously



Fig 1. Example with death rate 0.8. One cell division "detected" by change in population size actually requires five "real" cell divisions. Each of these "hidden" DNA replications gives extra chances to acquire a mutation.

(Fig 2) using a simulation approach. For an arbitrary chosen value of mutation rate toward a neutral arbitrary phenotype, we simulate the growth of a population of bacteria inoculated from a small number of nonmutant cells with a chosen constant death rate and track the number of mutant and nonmutant cells. We then compute the mutation rate based on the final state of these simulations, using the standard approach (i.e., the fluctuation test as described in the <u>Materials and methods</u>) to test whether we recover the true value of the mutation rate. As shown in Fig 2, the mutation rate is systematically overestimated when there is death, and the higher the death rate, the higher the overestimation. This result is unchanged when varying other population growth parameters, such as the initial and final population sizes, the mutation rate, and the plating fraction (underlying data have been uploaded to Zenodo 10.5281/ zenodo.1211765).



Fig 2. Overestimation of mutation rate when there is death. From simulations of population growth with known death and mutation rate, we estimate the mutation rate using the classical method, which does not take death into account. For each death rate between 0 and 0.95, 1,000 simulations with 24 parallel cultures were performed. For each simulation, we plot the ratio between the computed mutation rate (based on the number of mutants in the final state of the simulations) and the true mutation rate (used as input of the simulations, here $1*10^{-9}$ per division). The red lines indicate the median values, the boxes indicate the upper and lower quartiles, and the vertical bars indicate the upper and lower 5 percentiles. Underlying data have been uploaded to Zenodo (10.5281/zenodo.1211765).

Population dynamics and death in sub-MIC treatments

In the previous section, we show that it is necessary to take death into account when computing mutation rate. For this, tracking population size (and thus net growth rate) during antibiotic treatment, as classically done by plating and counting colony forming units, is not sufficient. It is not possible to know whether a decreased net growth rate in the treatment compared to the untreated control is due to a purely bacteriostatic effect (i.e., the population grows more slowly, but without death) or to a bactericidal effect (i.e., the bacteria keep dividing, potentially at the same rate as without antibiotic, but also die). The first scenario will have no effect on the accumulation of mutants as a function of population size, while in the second scenario, turnover implies a higher number of DNA replications and thus more mutants for a given population size, as explained above.

To disentangle these two effects, we designed a method allowing us to compute growth rate and death rate simultaneously using a segregative plasmid. The segregation dynamic allows us to estimate the number of bacterial cell divisions. Combining this information with the change in population size allows us to estimate growth rate and death rate, as explained in the <u>Materials and methods</u>.

Our ultimate goal is to reliably estimate mutation rates of bacteria treated with subinhibitory doses of antimicrobials. To this end, we quantify population dynamics and compute mutation rates toward a chosen neutral phenotype (resistance to rifampicin, conferred by substitutions in the gene rpoB) in populations exposed to subinhibitory doses of other antimicrobials. Our mutagenesis protocol is inspired by the standard fluctuation test with additional measurements of plasmid segregation to compute death rate, as detailed in the <u>Materials and</u> <u>methods</u>. The population dynamics are quantified as a combination of two variables: CFU at various time points (e.g., 0 h, 3 h, 6 h, and 24 h after treatment starts) and relative death rate (compared to birth rate) between pairs of two successive time points. We represent these population dynamics in <u>Fig 3</u> for the chosen subinhibitory antimicrobial treatments. We use kanamycin at 3 ug/mL, norfloxacin at 50 ng/mL, and hydrogen peroxide (H₂O₂) at 1 mM, allowing direct quantitative comparisons with the data from Kohanski and colleagues [28]. We also include untreated populations, in which we assume there is no death, to fit the plasmid segregation parameters (see <u>Materials and methods</u>).

We find that for norfloxacin, there is a strong death rate in all phases of growth and a strong impact of the treatment on final population size. For H_2O_2 , death is only detectable in stationary phase and the treatment is mostly bacteriostatic during growth. For kanamycin, the dynamics are more complex, because an initially high death rate leads to a strong decline of population size during the first 6 h of growth, followed by a recovery leading to a final population size close to the one reached in untreated controls. During this second phase of growth following the bottleneck at 6 h, death rate is still substantial. This clearly shows that none of the three studied treatments are fully sublethal and thus that the implicit assumption of no death made when using the standard methods of computation of mutation rate (as done by Kohanski and colleagues [28]) does not apply.

Mutation rate in sub-MIC treatments

We developed computational tools to quantify mutation rate, taking into account the measured population dynamics and accounting for death. Our software, ATREYU (Approximate bayesian computing Tentative mutation Rate Estimator that You could Use), is described in the <u>Materials and methods</u>. It takes as input any arbitrary population dynamics, described as a list of population sizes (i.e., CFU/mL for several time points) and an associated list of death rates between pairs of consecutive time points. This input is thus exactly what is shown in





Fig 3. Growth and death dynamics of populations treated with sub-MIC antimicrobials. Each panel shows data for treatment with a different antimicrobial. Blue dots (left axis) joined by straight blue line represent population sizes measured by plating, expressed as CFUs per mL of culture. Red line (right axis) represents relative death rates, computed from plasmid segregation data, corresponding to the average number of death event per division event between two successive time points. For each treatment, the behaviors of at least 4 fully independent biological replicates, performed on different days with different batches of medium and comprising at least 4 replicate populations, are averaged. Error bars indicate standard error of the mean between biological replicates. Death rates higher than 5 were set to 5. See also <u>S1</u> <u>Supporting Information</u> for a representation of all biological replicates. Underlying data have been uploaded to Zenodo (10.5281/zenodo.1211765). CFU, colony-forming unit; MIC, minimal inhibitory concentration.

Fig 3. We apply this method to analyze the results of our mutagenesis protocol, to quantify whether and by how much subinhibitory treatments with kanamycin, norfloxacin, or H_2O_2 increase mutation rate. We show the effect of treatment on mutation rate in Fig 4. We also plot the uncorrected mutation rate estimate, assuming no death as would be obtained by methods such as FALCOR (Fluctuation AnaLysis CalculatOR) [39], bzRates [40], or rSalvador [38]. Clearly, not taking death into account leads to a strong overestimation of the mutation rate for both kanamycin and norfloxacin. In the case of kanamycin, correctly computing the mutation rate removes all signal for stress-induced mutagenesis. In the case of norfloxacin, this signal is strongly lowered, from a 14-fold to a 6-fold increase. For H_2O_2 , the signal is less affected, which can be attributed to death rate only being significant in stationary phase. This confirms



Fig 4. Mutation rate when treating with a sub-MIC dose of kanamycin, H₂O₂, or norfloxacin compared to untreated. The uncorrected mutation rate is the one that would be computed based on our data when death is not taken into account. Each point corresponds to a fully independent biological replicate comprising 24 parallel populations. All the computed mutation rates are normalized by the average mutation rate computed in absence of treatment. The horizontal dashed line (y = 1) thus indicates the mutation rate of untreated populations. For each treatment, we performed a paired t test to estimate whether the corrected mutation rate was significantly lower than the uncorrected one (* p < 0.05, ** p < 0.01). Underlying data have been uploaded to Zenodo (10.5281/ zenodo.1211765).

that neglecting death leads to a systematic overestimation of mutation rates and that taking into account the full population dynamics is necessary and leads to significantly different patterns depending on the antimicrobial and its effect on growth and death.

The link between evolvability and mutation rate depends on population dynamics

The quantification of mutation rate in different conditions is not sufficient to answer the question of whether subinhibitory antibiotic treatments increase evolvability in general and, in particular, increase the likelihood of emergence of a resistant mutant and thus the probability of treatment failure. Indeed, mutation rate is expressed per DNA division, but, as we have shown in the previous section, antibiotic treatment may significantly change the number of susceptible cells and the number of replications that these cells have undergone. Intuitively, if a treatment multiplies mutation rate by 10 but divides population size by 100, it is not likely to lead to an increased genetic diversity. This intuition has also been given by Couce and Blázquez (Fig 2 of [45]) but has been largely ignored in the literature as it was not the main message of this review. Conversely, a treatment that does not affect mutation rate and only slightly affects carrying capacity but causes death and turnover may result in a significantly increased genetic diversity.

We first show the effect of subinhibitory treatment on final population size in Fig 5. While H_2O_2 does not affect final population size, there is a strong effect of 1–2 orders of magnitude



Fig 5. Population size reached after 24 h in treated and untreated conditions. Each point corresponds to a fully independent biological replicate. For each of these biological replicates, average population size is estimated by plating appropriate dilutions of at least 6 replicate populations on nonselective medium. The dashed black line indicates the average size of the untreated populations. We performed an unpaired *t* test to estimate whether treated population sizes are significantly different than untreated ones (** p < 0.01). Underlying data have been uploaded to Zenodo (10.5281/zenodo.1211765).

for norfloxacin and a significant but smaller effect of around 50% reduction for kanamycin. This supports our intuition that at least for norfloxacin, the few-fold increase in mutation rate we report (Fig 4) is probably uncorrelated to any increase in genetic diversity.

We expect the generation of genetic diversity to depend on (i) the number of cells alive, (ii) the population dynamics of these cells, and (iii) their mutation rate. Addressing the effect of stress on mutation rate as done in the previous section is necessary for a proper understanding of the bacterial stress response and of DNA repair mechanisms. Nevertheless, mutation rate is not the relevant measure to understand the effect of stress on the generation of genetic diversity and thus on evolvability.

As a simple quantification of the generation of genetic diversity and thus an approximation of evolvability, we measure the number of mutants at a neutral locus, here the base-pair substitutions conferring resistance to rifampicin in the gene rpoB.

We plot in Fig.6 the absolute number of rifampicin-resistant mutants in the final population for all treatments and for untreated control. Evolvability is reduced by a few-fold by kanamycin treatment (as expected, since this treatment decreases population size without increasing mutation rate). While norfloxacin and H_2O_2 both induce a small increase in mutation rate, they interestingly have strongly opposite effects on evolvability. Treatment with H_2O_2 increases evolvability by more than one order of magnitude, while treatment with norfloxacin reduces it by a similar amount. This is due to the very different effects these antimicrobials have on population dynamics: while H_2O_2 does not affect final population size, norfloxacin causes a strong decrease in population size due to both bactericidal and bacteriostatic effects.



Fig 6. Evolvability of untreated and treated populations. For each treatment, we estimate evolvability as the number of rifampicin-resistant mutants in the population after 24 h of growth. The dashed black line represents the average evolvability of the untreated populations. Each point is a fully independent biological replicate comprising 24 replicate populations in which the number of rifampicin-resistant mutants is averaged. We performed an unpaired *t* test to estimate whether evolvability of treated populations is significantly different than this of untreated ones (** p < 0.01). Underlying data have been uploaded to Zenodo (10.5281/zenodo.1211765).

https://doi.org/10.1371/journal.pbio.2005056.g006

So independently of the question whether antibiotics increase mutation rate, we show that the sub-MIC treatments we studied do not in any way increase evolvability. Thus, the standard rationale, that these subinhibitory treatments would increase the risk of emergence of resistance and treatment failure because of a higher generation of genetic diversity [46], does not hold.

This effect is largely due to a strong reduction in population size, which implies a loss in genetic diversity. Population size and mutation rate are not, however, the only factors affecting evolvability. We may also ask how much the measured turnover in our experiments contributes to evolvability. To answer this question, we simulate the same population dynamics as observed for each treatment but without death: Each population reaches the same final population size as measured in our experiments, with the same mutation rate as computed, but with no death. This is similar to what would happen if the antibiotics only had a bacteriostatic effect. For each simulation, we quantify evolvability using the same measure as previously, i.e., the absolute number of mutants for our phenotype of interest in the final population. We compare this simulated evolvability without turnover with the actual measured evolvability in Fig 7. For kanamycin and norfloxacin, turnover significantly increases evolvability by a few-fold.

Conclusions

In summary, our results show that (1) mutation rate is systematically overestimated in subinhibitory treatments because of death, (2) mutation rate is not the only parameter that controls

LOS BIOLOGY



Fig 7. Contribution of turnover to evolvability. For each treatment, we plot the measured evolvability of the populations (red triangles, purple dots, and blue squares) and the estimate that was obtained if the same population size was attained without cell death and with the same mutation rate (violin plots, 100 replicate simulations for each biological replicate). The median of all simulations for a given treatment is represented as a horizontal bar. For each treatment, we performed an unpaired *t* test to test whether the evolvability with death is different than the evolvability without death (** p < 0.01). Underlying data have been uploaded to Zenodo (10.5281/zenodo.1211765).

https://doi.org/10.1371/journal.pbio.2005056.g007

the generation of genetic diversity or evolvability, (3) population size and turnover play a key role in evolvability, and (4) treatment with subinhibitory doses of norfloxacin or kanamycin significantly decreases evolvability, measured as the generation of genetic diversity at population scale. These results are in apparent disagreement with the conclusions of previous studies on antibiotic-induced mutagenesis. This discrepancy is due to both miscalculation of mutation rates (which occurs when one neglects the effect of population dynamics) and misconceptions about the link between mutation rate and evolvability in these classical papers.

Discussion

Understanding genetic and environmental control of evolvability is central for the understanding of microbial adaptation to constantly changing environments. Evolvability is defined as the capacity of a population to generate adaptive genetic diversity. This can be decomposed in two variables: the amount of genetic diversity generated by a population (often inaccurately attributed to the mutation or recombination rate only) and the fraction of this diversity that is adaptive. We are here interested in the former. Genetic control over the amount of generated genetic diversity has been studied for a long time in the field of mutation rate evolution [20,22]. The existence of constitutive mutator alleles in bacteria has been discovered before the mechanisms of DNA replication [47], and the selection pressures leading to their transient increase in frequency have been elucidated through both theoretical and experimental studies [23,48]. Observing the evolution and fixation of such mutator alleles from nonmutator lineages in a long-term evolution experiment [49] plausibly facilitated the acceptance of these theories. On the other hand, plastic, environment-dependent control over the generation of genetic diversity has been a controversial paradigm shift in bacterial evolution [18].

It has been proposed for a long time that various stresses can increase mutation rates in bacteria [9,10], including those triggered by antimicrobial treatments [28-32]. Several molecular pathways have been shown to be implicated in this phenomenon, the emblematic one being the SOS response [12]. In this work, we have shown that the effect of stress on mutation rate can not be computed properly with the existing tools, because the underlying mathematical models make the assumption that there is no stress or, more precisely, that the stress does not affect population dynamics. We develop experimental and computational tools to measure population dynamics and compute mutation rates under stress and apply them to the question of mutagenesis due to antibiotic treatment. We have shown that the intuition that low doses of antibiotics are dangerous because they lead to a higher generation of diversity is based on a misinterpretation of valid experimental data for two reasons: (1) the increase in mutation rate is overestimated due to overly simplistic assumptions, and (2) a higher mutation rate does not lead to a higher generatic diversity if population dynamics are affected (e.g., if population size is reduced).

The question of emergence of resistance alleles due to low doses of antibiotics (reviewed by Andersson and Hughes [50]) cannot, however, be entirely addressed by measuring the generation of genetic diversity. The study of adaptive evolution can be decomposed in two parts: generation of diversity and natural selection acting on this diversity. While we have shown that treatment with a subinhibitory dose of norfloxacin does not increase but rather strongly decreases the amount of generated genetic diversity, it has also been reported that resistance alleles can be maintained and enriched by selection, even at very low antibiotic concentration [51]. Such selection of preexistent alleles may be a much more valid reason for concern about subinhibitory treatments. However, the literature is not as unanimous regarding bacteria residing within a patient with an immune system, rather than in a test tube [52]. It has, for example, been suggested that treating with a lower dose of antibiotics could slow down the selection of existing resistance alleles by decreasing their fitness advantage compared to the sensitive, wild-type strain, without compromising the success of the treatment [53, 54]. Combining our results with these papers calls for a reevaluation of the evolution of antibiotic resistance at low doses of antibiotics.

The question of the potentially adverse effects of low doses of antibiotics has been of longstanding interest in the medical community, as is evidenced by the famous quote from Alexander Fleming's Nobel lecture [55], "If you use penicillin, use enough." However, given the time of this research (penicillin was discovered in 1928 and thus 15 years before Luria and Delbrück), one should not be surprised that this often cited out-of-context advice relies on a rather Lamarckian reasoning in terms of educating rather than selecting for resistance:

Then there is the danger that the ignorant man may easily underdose himself and by exposing his microbes to non-lethal quantities of the drug make them resistant. Here is a hypothetical illustration. Mr. X. has a sore throat. He buys some penicillin and gives himself, not enough to kill the streptococci but enough to educate them to resist penicillin...

Our findings are also relevant outside of the context of evolution during antibiotic treatment. As we mentioned, mutagenesis in bacteria under nutritional stress was a key development in the understanding of the bacterial stress response and DNA repair, with a recent regain of interest [7,8]. Our experimental system can a priori not be applied to study starving bacteria, for two reasons: (1) our plasmid segregation method only gives sufficient signal in nonstationary populations, and (2) many of the observations on starvation-induced mutagenesis are dependent on the presence of some spatial structure (for example, bacterial colonies on agar plates [2,9,10]). In this second case, the population dynamics become much more complex and are unlikely to be realistically approximated by a single relative death rate parameter. But the exact same questions remain to be elucidated in this field: How many cell divisions happen in these starving colonies? In batch cultures, is stationary phase really stationary, or is there some turnover and recycling as recently suggested [56]? And more importantly, where does death come from: is it an unavoidable, externally caused phenomenon; or is there an internal component, such as an altruistic programmed cell death [57], or just traits selected in other environments that give a maladaptation to certain stresses [58]?

Stress-induced mutagenesis is of interest for several research fields, with different questions. We showed that the relevant question in a clinical setting is not directly about mutation rate but about evolvability and that the link between both is confunded by the effect of treatment on population dynamics.

One the evolutionary side, the central question about stress-induced mutagenesis is "Is it adaptive?" Studying the molecular mechanisms of stress response will shed light on one part of the answer: is the increase in mutation rate controlled by the cell, or is it an unavoidable consequence of the stress? In this regard, the three antimicrobials we study seemingly have very different properties. H_2O_2 is creating reactive oxygen species that directly damage DNA independently of the cell machinery, iron being the only necessary catalyst [43,44]. The way DNA damage leads to mutations is controlled by the cell but is more likely to be a consequence of selection for survival ("survival at the price of the mutation"), rather than selection for evolvability. On the other hand, kanamycin acts on protein synthesis [41], and any hypothetical mutagenic effect would thus go through the cell machinery. Norfloxacin is in between, because it acts on the DNA-gyrase complex, leading to an arrest of DNA synthesis and, in some conditions, to double strand breaks [42].

Recent findings from J. Collins and colleagues, however, suggest that these different scenarios are not as distant as they may seem, because they suggest that the production of reactive oxygen species is a feature of all bactericidal antibiotics [33–35]. While supporting the idea that antibiotic treatment increases mutation rate and does so in correlation with bactericidal activity, these debated findings would also suggest that such increase in mutation rate does not stem from selection for evolvability.

In a nutritional stress scenario, Maharjan and colleagues [8] show that at equal effect on growth rate, limitation of different nutrients has very different effects not only on mutation rate but also on mutational spectrum, again showing the need for a mechanistic understanding of the molecular details and suggesting that the evolutionary outcome is much more complex than a linear increase in mutation rate in response to starvation.

We provide tools that may help further developments of these questions. Our software, ATREYU, can be used to compute mutation rates from mutant counts in populations with arbitrary but known birth and death dynamics. The mutant counts are obtained by a protocol similar to the classical fluctuation test. The birth and death dynamics can be obtained by several methods. We used plasmid segregation, but other methods may be possible, such as segregation of engineered self-assembling fluorescent particles [59], isogenic strain tagging [60], Carboxyfluorescein succinimidyl ester (CFSE) membrane staining [61], or direct microscopic observations at single-cell resolution. Microscopic observations with cell tracking may give much more precise and less noisy information than other methods but are only suitable when the death rate is sufficiently low, because only a limited number of cells can be tracked. We believe that death and cell turnover are crucial factors in evolutionary microbiology but are often neglected, in part due to the lack of standard methods to measure them. In immunology, in which the population dynamic of lymphocytes has been recognized as a central question

[62], many methods have been developed [63], including the aforementioned CFSE membrane staining.

While our method could be adapted for many nonstandard assumptions other than death (e.g., fitness cost of the mutation or partial plating), it shares some of the limitations of the more classical systems. Firstly, we suppose that each cell is fully monoploid and has only one chromosome and thus that the number of DNA replications is the number of cell divisions. However, some quinolone antibiotics are known to cause filamentation [64], increasing the number of chromosomes per cell [42] and potentially changing the evolutionary dynamics [65]. Further complicating the picture, recent work [66] shows that even within a single chromosome, multifork replication may cause different ploidy levels on different loci, affecting mutation rate estimates and evolutionary dynamics. Secondly, we also consider that time does not matter, in the sense that the probability of mutation per division is independent of the growth rate, and that nondividing bacteria do not accumulate mutations, justifying the expression of mutation rate as a quantity of mutations per division event and not per unit of time. Since Luria's and Delbrück's experiment, this has been the standard assumption both on the microbiological and mathematical side [67]. However, recent data on fission yeasts suggest that nonreplicating cells may accumulate mutations at a different rate and spectrum than diving cells [68]. Finally, we make the assumption of homogenous behavior in the population, excluding the possibility that different subpopulations have different death and mutation rates. The question of whether a small subpopulation in a different physiological state may contribute most of the mutational supply is still unresolved. Theoretical work [69] shows that such situation could have a large impact on the evolutionary dynamics.

Zooming out from evolutionary microbiology, mutagenesis research in bacteria shows an interesting parallel with recent advances in cancer research. For a given cell growth dynamic (organogenesis, from stem cells to an organized population of differentiated cells), a higher mutation rate (expressed per cell division) will boost the accumulation of mutations and thus the risks of cancer. This increase in mutation rate can be genetic, such as in the case of hereditary nonpolyposis colon cancer caused by a deficiency of mismatch repair [70], or environmental, such as exposure to carcinogenic compounds [71–73]. All of this is now part of textbook science on cancer and is similar to an increase in mutation rate in a bacterial population due to genetic (mutator alleles [47]) or environmental (stress-induced [18] or stress-associated [26] mutagenesis) factors.

Tomasetti and Vogelstein [74] recently reported that the number of stem cell divisions is a strong predictor of cancer risk per organ. This is in parallel with our findings, which show that the number of cell divisions is central to predict the generated genetic diversity in a population of cells. Tomasetti and Vogelstein caused a major controversy by concluding that cancers would thus mostly be due to "bad luck" (i.e., unavoidable consequences of the large number of cell divisions) rather than to environmental factors (e.g., exposure to mutagenic chemicals). We show here that the generation of genetic diversity depends on both mutation rate and cell population dynamic, which is in line with many studies that have criticized the interpretation of the data made by Tomasetti and Vogelstein.

The challenge of understanding evolvability in bacterial population is thus strikingly similar to the one of understanding cancer, in the sense that the outcome depends on a complex interplay of extrinsic and intrinsic factors acting at different scales. In the case of bacteria, additional complexity stems from the fact that the same treatments may both impact the number of cell divisions (death and turnover) and the mutagenicity of each division. The picture is further complicated by the difficulty of disentangling the direct effects of the drug from the effects of the stress response triggered by the drug. But fortunately, while separating and measuring each factor requires complex experimental methods and mathematical tools, measuring evolvability on neutral loci is simpler, at least in bacteria. We hope that our study will encourage researchers in the field to question more not only the appropriateness of the tools they use for mutation rate estimation and the assumptions implicitly made by using these tools but also the pertinence of the variable they choose to report.

Materials and methods

Experimental setup

Our mutagenesis protocol is directly inspired by the one used by Kohanski and colleagues [28] (which is in turn similar to that of Luria and Delbrück [75]) with the inclusion of a segregative plasmid to compute death rate, as explained further below and graphically represented on Fig.8.

A culture of *E. coli MG1655* (with plasmid pAM34) is inoculated from a freezer stock and grown overnight in LB supplemented with 0.1 mM IPTG and 100 ug/mL of ampicillin (to ensure maintenance of pAM34). After the culture reaches stationary phase (at least 15 h of growth), it is washed 3 times in normal saline (9 g/L NaCl) to remove traces of IPTG and then diluted 10,000 times in a 500 mL baffled flask containing 50 mL of LB (to maximize oxygenation). After 3.5 h of growth, the culture is inoculated at a ratio of 1:3 in 24 culture tubes containing a total volume of 1 mL of LB supplemented with one of the studied antimicrobials at subinhibitory concentration (3 ug/mL kanamycin, 50 ng/mL norfloxacin, 1 mM hydrogen peroxide, or untreated control). After 24 h of growth at 37°C, the cultures are plated at appropriate dilutions on 3 different LB agar medium: LB only to count the total number of bacteria (CFU), LB supplemented with 100 ug/mL ampicillin + 0.1 mM IPTG to count the number of bacteria bearing a copy of the segregative plasmid, and LB supplemented with 100 ug/mL rifampicin (plated volume: 200 uL) to count the number of mutants toward the phenotype of interest. Additionally to this 24 h time point, cultures are also plated on LB and LB + ampicillin



+ IPTG at intermediate time points (3 h and 6 h) to have a more accurate quantification of plasmid segregation dynamics and thus a better time resolution for the estimation of death rate. The plates are incubated between 15 h and 24 h for LB and LB ampicillin IPTG and exactly 48 h for LB rifampicin before counting colonies. Further experimental details are given in <u>S5 Supporting information</u>.

Measuring death using plasmid segregation

pAM34 is a colE1 derivative whose replication depends on a primer RNA put under the control of the inducible promoter pLac [76]. Under the presence of 0.1-1mM IPTG (nonmetabolizable inducer of the lactose operon), the plasmid is stably maintained in every cell. When IPTG is removed from the growth medium, the plasmid is not replicating anymore, or not as fast as the cells divide, and thus is stochastically segregated at cell division. The decrease in plasmid frequency between two time points then allows us to compute the number of bacterial cell divisions that occurred between these two time points. Combined with the change in population size, this allows us to compute average death rate and growth rate between these two time points (see Fig.9 and mathematical explanations below). Such segregation measures have been used in a less quantitative way by other researchers [77,78] to crudely infer overall population turnover in vivo.

pAM34 also carries a betalactamase. The number of plasmid-bearing bacteria can thus be counted by plating an appropriate dilution of the culture on LB supplemented with 0.1 mM IPTG (to ensure maintenance of the plasmid within colonies founded by a plasmid-bearing cell) and 100 ug/mL ampicillin (to only permit growth of colonies founded by a plasmid-bearing cell). The total number of bacteria is determined by plating an appropriate dilution of the culture on LB.

Because mutational dynamic does not depend on time, we chose to compute relative death rate (ratio of death rate and growth rate as functions of time), which is the average number of death events per division event.

The link between plasmid segregation, death, and number of divisions between two time points can be expressed mathematically as follows.

If we have the following:

- F: the frequency of cells bearing at least a copy of the plasmid, measured by plating on LB + ampicillin + IPTG;
- *N*: the total number of cells, measured by plating on LB;
- *res*: the rate of residual replication of the plasmid relative to the division rate in absence of IPTG;



Fig 9. Segregation of a plasmid with inducible replication allows us to estimate death rate. (a) Maintenance of the plasmid in presence of IPTG, (b) segregation of the plasmid in absence of IPTG, (c) computing population dynamics (death rate) from plasmid segregation in absence of IPTG. IPTG, Isopropyl β-D-1-thiogalactopyranoside.

- *g*: the number of generations, i.e., the average number of duplications each genome present at final time did undergo; and
- *d*: the relative death rate (temporal death rate divided by temporal division rate);

then the plasmid is diluted/segregated at each division following the equation

$$F_{final} = F_{initial} \times (\frac{1 + res}{2})^g$$

So we can estimate

$$g = log_2(\frac{F_{final}}{F_{initial}})/log_2(\frac{1 + res}{2})$$

Without any death, we would have

$$g_{no-death} = log_2(N_{final}/N_{initial})$$

The difference between the true number of generations *g* computed from plasmid frequency and this number of generations $g_{no-death}$ computed based on the assumption that there is no death, allows us to estimate relative death rate as follows:

$$N_{\text{final}} = N_{\text{initial}} \times 2^{(1-d)*g}$$

This yields

$$d = 1 - rac{log_2(N_{final}/N_{initial})}{g}$$

and thus

$$d = 1 - rac{log_2(N_{final}/N_{initial})}{log_2(F_{final}/F_{initial})} imes log_2(rac{1 + \mathit{res}}{2})$$

The only remaining free parameter to estimate is *res*, which is estimated by performing growth kinetics without antibiotic treatment (in LB medium) and thus without (or with negligible) death. We then have

$$g = g_{no-death}$$

and thus

$$log_{2}(\frac{1 + res}{2}) = \frac{log_{2}(F_{final}/F_{initial})}{log_{2}(N_{final}/N_{initial})}$$

from which we can fit the value of the segregation parameter $log_2(\frac{1+res}{2})$ based on the values of *F* and *N* estimated by plating. Further experimental and mathematical details on the plasmid segregation system are given in <u>S2–S5</u> Supporting Informations.

Computing mutation rate taking death into account

Most modern measures of mutation rate rely on the same standard protocol, the fluctuation test [79], directly inspired by the Luria and Delbrück experiment [75]: several cultures are inoculated with a small population of nonmutant bacteria, are grown overnight and are then plated on selective media (to count the number of mutants in the final population) and on nonselective media (to count the total number of bacteria in the final population). The number

of mutants *r* in the final population (or, rather, its distribution over several replicate populations) is used to estimate the number of mutational events *m* happening during growth. One should note that these two numbers are not equivalent, because one mutational event can lead to several mutants in the final population if it happens early during growth, making this part of the computation complicated for intuition, although good mathematical tools are available. The total number of bacteria *N* is assumed to be very close to the number of cell divisions (and thus the number of genome replications) because the initial number of bacteria is much smaller. Mutation rate can thus be estimated as $\mu = m/N$.

The many existing software packages used to compute *m* from the observed distribution of *r* use an analytical expression of the probability-generating function (pgf) of the number of mutants in the final population [80]. The only free parameter is the number of mutational events (equivalent to the value of the mutation rate per division when scaled with population size). This parameter is estimated from plating data using the maximum likelihood principle. The most used implementation of this idea is FALCOR [39], available on a webpage: <u>http://www.keshavsingh.org/protocols/FALCOR.html</u>.

Other software packages implementing the same ideas have been developed more recently, including, for example, rSalvador [38] and bzRates [40], which also implement a few alternative assumptions such as fitness impact (cost or benefit) of the focal mutation or a more accurate correction for plating efficiency than the one suggested by FALCOR [81].

However, to this day, no available software allows users to compute mutation rate when there is death. Some papers derived analytical expression of the pgf of the number of mutants in the final population in conditions in which there is death [82], but this has to our knowledge never been applied to real data nor implemented in a software package. In theory, such computations could easily be implemented in a tool similar to FALCOR (web server) or rSalvador (software package). However, the basic assumption of the derived formula is that death rate is constant. This assumption is the price to pay for an analytical expression for the pgf and is unfortunately not appropriate in our case, given the observed death kinetics (see Fig 3). On the other hand, given the computational power available today, we believe that analytical computations are not always necessary. In our case, while the measured population dynamics do not allow us to derive an analytical expression of the pgf, it is straightforward to simulate many times such population dynamics with an arbitrary mutation rate and to obtain an empirical distribution of the number of mutants. Running these simulations for any possible value of the mutation rate parameter then allows Bayesian inference: we look for the simulated mutation rate that gives the closest distribution to the one experimentally observed, as graphically represented in Fig 10. Such methods are classically referred to as Approximate Bayesian Computing.



From the observed population dynamics (CFU and death):

Simulate population growth with many μ (10,000 replicate simulations for each value)

Which µ best matches experimental data?

Fig 10. Using ATREYU to compute mutation rate from mutant counts and population dynamics. ATREYU takes as inputs the observed population dynamics (CFU counts at several time points and death rates between pairs of successive time points) and the count of mutants toward the phenotype of interest in several replicate cultures. Because it is based on simulation and does not require analytical expression of the mutant distribution, any nonstandard biological assumption (death, in our case) can easily be implemented. ATREYU, Approximate bayesian computing Tentative mutation Rate Estimator that You could Use; CFU, colony-forming unit.

We implemented such simulations and inference in a Python software package, ATREYU, and use this software as the heart of our data analysis.

Supporting information

S1 Supporting Information. Detailed visualization of population dynamics for all biological replicates. Figure equivalent to Fig.3 of the main text, but with the behavior all biological replicates analyzed and represented independently. (PDF)

S2 Supporting Information. Measuring plasmid frequency: Mathematical assumptions and technical details. Experimental details on the use of the plasmid pAM34: dealing with plasmid copy number, appropriate quantity of IPTG, no satellite colonies observed. (PDF)

S3 Supporting Information. Testing the plasmid segregation approach to compute death rate. Estimating the accuracy of our estimation of death rate, using experiments in which death is mimicked with dilutions. (PDF)

S4 Supporting Information. Norfloxacin does not prevent plasmid replication in presence of the inducer. Norfloxacin is not likely to act directly on plasmid replication, because it does not cause plasmid segregation in presence of the inducer. (PDF)

S5 Supporting Information. Preparation of medium, strains, and reactants, and additional experimental details. Other experimental details: preparation of the strain, minimizing the variability between biological replicates by careful preparation and storage of reactants, behavior of the plasmid in a Δ lacY mutant and in presence of glucose. (PDF)

Acknowledgments

We express our gratitude to Balazs Bogos and Arnaud Gutierrez for their comments on an earlier version of this manuscript; to the members of our group, and especially Julien Barrere, for many discussions; to Sandra Wotzka and Emma Slack for providing the plasmid pAM34; and to Chantal Lotton and Ivan Matic for providing temporary lab space, allowing us to perform an extra experiment after the first author changed labs. The computing power needed by this study was provided by the High-Performance Computing team of ETH Zürich Scientific IT Services.

Author Contributions

Conceptualization: Antoine Frenoy.

Funding acquisition: Sebastian Bonhoeffer.

Investigation: Antoine Frenoy.

Writing - original draft: Antoine Frenoy, Sebastian Bonhoeffer.

References

Cairns J, Overbaugh J, Miller S. The origin of mutants. Nature. 1988; 335:142–145. <u>https://doi.org/10.1038/335142a0</u> PMID: 3045565

- Cairns J, Foster PL. Adaptive reversion of a frameshift mutation in *Escherichia coli*. Genetics. 1991; 128:695–701. PMID: <u>1916241</u>
- Stahl FW. A unicorn in the garden. Nature. 1988; 335:112–113. <u>https://doi.org/10.1038/335112a0</u> PMID: <u>3412467</u>
- 4. Torkelson J, Harris RS, Lombardo MJ, Nagendran J, Thulin C, Rosenberg SM. Genome-wide hypermutation in a subpopulation of stationary-phase cells underlies recombination-dependent adaptive mutation. EMBO J. 1997; 16:3303–3311. <u>https://doi.org/10.1093/emboj/16.11.3303</u> PMID: <u>9214645</u>
- Andersson DI, Slechta ES, Roth JR. Evidence that gene amplification underlies adaptive mutability of the bacterial *lac* operon. Science. 1998; 282:1133–1135. <u>https://doi.org/10.1126/science.282.5391.</u> <u>1133</u> PMID: <u>9804552</u>
- Sano E, Maisnier-Patin S, Aboubechara JP, Quiñones-Soto S, Roth JR. Plasmid copy number underlies adaptive mutability in bacteria. Genetics. 2014; 198:919–933. <u>https://doi.org/10.1534/genetics.114.</u> <u>170068</u> PMID: <u>25173846</u>
- Krašovec R, Belavkin RV, Aston JAD, Channon A, Aston E, Rash BM, et al. Mutation rate plasticity in rifampicin resistance depends on *Escherichia coli* cell-cell interactions. Nat Commun. 2014; 5:3742. <u>https://doi.org/10.1038/ncomms4742</u> PMID: 24776982
- Maharjan RP, Ferenci T. A shifting mutational landscape in 6 nutritional states: Stress-induced mutagenesis as a series of distinct stress input–mutation output relationships. PLoS Biol. 2017; 15: e2001477. https://doi.org/10.1371/journal.pbio.2001477 PMID: 28594817
- Taddei F, Matic I, Radman M. cAMP-dependent SOS induction and mutagenesis in resting bacterial populations. Proc Natl Acad Sci USA. 1995; 92:11736–11740. <u>https://doi.org/10.1073/pnas.92.25.</u> <u>11736</u> PMID: <u>8524839</u>
- Bjedov I, Tenaillon O, Gerard B, Souza V, Denamur E, Radman M, et al. Stress-induced mutagenesis in bacteria. Science. 2003; 300:1404–1409. <u>https://doi.org/10.1126/science.1082240</u> PMID: <u>12775833</u>
- 11. Wrande M, Roth JR, Hughes D. Accumulation of mutants in "aging" bacterial colonies is due to growth under selection, not stress-induced mutagenesis. Proc Natl Acad Sci USA. 2008; 105:11863–11868. https://doi.org/10.1073/pnas.0804739105 PMID: 18701713
- Radman M. SOS repair hypothesis: phenomenology of an inducible DNA repair which is accompanied by mutagenesis. In: Molecular mechanisms for repair of DNA. Springer; 1974. p. 355–367. PMID: 1103845
- 13. Schlacher K, Goodman MF. Lessons from 50 years of SOS DNA-damage-induced mutagenesis. Nat Rev Mol Cell Biol. 2007; 8:587–594. <u>https://doi.org/10.1038/nrm2198</u> PMID: <u>17551516</u>
- Courcelle J, Khodursky A, Peter B, Brown PO, Hanawalt PC. Comparative gene expression profiles following UV exposure in wild-type and SOS-deficient *Escherichia coli*. Genetics. 2001; 158:41–64. PMID: <u>11333217</u>
- Tippin B, Pham P, Goodman MF. Error-prone replication for better or worse. Trends Microbiol. 2004; 12:288–295. https://doi.org/10.1016/j.tim.2004.04.004 PMID: 15165607
- Wagner J, Gruz P, Kim SR, Yamada M, Matsui K, Fuchs RPP, et al. The *dinB* gene encodes a novel *E. coli* DNA polymerase, DNA Pol IV, involved in mutagenesis. Mol Cell. 1999; 4:281–286. <u>https://doi.org/10.1016/s1097-2765(00)80376-7</u> PMID: 10488344
- Tang M, Pham P, Shen X, Taylor JS, O'Donnell M, Woodgate R, et al. Roles of *E. coli* DNA polymerases IV and V in lesion-targeted and untargeted SOS mutagenesis. Nature. 2000; 404:1014–1018. <u>https://doi.org/10.1038/35010020</u> PMID: <u>10801133</u>
- Galhardo RS, Hastings PJ, Rosenberg SM. Mutation as a stress response and the regulation of evolvability. Crit Rev Biochem Mol Biol. 2007; 42:399–435. <u>https://doi.org/10.1080/10409230701648502</u> PMID: 17917874
- 19. Ram Y, Hadany L. The evolution of stress-induced hypermutation in asexual populations. Evolution. 2012; 66:2315–2328. https://doi.org/10.1111/j.1558-5646.2012.01576.x PMID: 22759304
- Tenaillon O, Taddei F, Radman M, Matic I. Second-order selection in bacterial evolution: Selection acting on mutation and recombination rates in the course of adaptation. Res Microbiol. 2001; 152:11–16. https://doi.org/10.1016/S0923-2508(00)01163-3 PMID: 11281320
- 21. Eshel I. Clone selection and the evolution of modifying features. Theor Popul Biol. 1973; 4:196–208. https://doi.org/10.1016/0040-5809(73)90029-4
- Chao L, Cox EC. Competition between high and low mutating strains of *Escherichia coli*. Evolution. 1983; 37:125–134. <u>https://doi.org/10.1111/j.1558-5646.1983.tb05521.x</u> PMID: <u>28568016</u>
- Taddei F, Radman M, Maynard Smith J, Toupance B, Gouyon PH, Godelle B. Role of mutator alleles in adaptive evolution. Nature. 1997; 387:700–702. <u>https://doi.org/10.1038/42696</u> PMID: <u>9192893</u>

- Chicurel M. Can organisms speed their own evolution? Science. 2001; 292:1824–1827. <u>https://doi.org/10.1126/science.292.5523.1824</u> PMID: <u>11397928</u>
- Singer E. Under pressure, does evolution evolve?; 2014. Available from: <u>https://www.quantamagazine.org/under-pressure-does-evolution-evolve-20140115/</u>.
- MacLean RC, Torres-Barceló, Moxon R. Evaluating evolutionary models of stress-induced mutagenesis in bacteria. Nat Rev Genet. 2013; 14:221–227. <u>https://doi.org/10.1038/nrg3415</u> PMID: 23400102
- Obolski U, Hadany L. Implications of stress-induced genetic variation for minimizing multidrug resistance in bacteria. BMC Med. 2012; 10:89. <u>https://doi.org/10.1186/1741-7015-10-89</u> PMID: <u>22889082</u>
- Kohanski MA, DePristo MA, Collins JJ. Sublethal antibiotic treatment leads to multidrug resistance via radical-induced mutagenesis. Mol Cell. 2010; 37:311–320. <u>https://doi.org/10.1016/j.molcel.2010.01.</u> 003 PMID: 20159551
- 29. Baharoglu Z, Mazel D. Vibrio cholerae triggers SOS and mutagenesis in response to a wide range of antibiotics: A route towards multiresistance. Antimicrob Agents Chemother. 2011; 55:2438–2441. https://doi.org/10.1128/aac.01549-10 PMID: 21300836
- Gutierrez A, Laureti L, Crussard S, Abida H, Rodríguez-Rojas A, Blázquez J, et al. B-Lactam antibiotics promote bacterial mutagenesis via an RpoS-mediated reduction in replication fidelity. Nat Commun. 2013; 4:1610. https://doi.org/10.1038/ncomms2607 PMID: 23511474
- Rodríguez-Rojas A, Makarova O, Rolff J. Antimicrobials, stress and mutagenesis. PLoS Pathog. 2014; 10: e1004445. <u>https://doi.org/10.1371/journal.ppat.1004445</u> PMID: 25299705
- Long H, Miller SF, Strauss C, Zhao C, Cheng L, Ye Z, et al. Antibiotic treatment enhances the genomewide mutation rate of target cells. Proc Natl Acad Sci USA. 2016; 113:E2498–E2505. <u>https://doi.org/10. 1073/pnas.1601208113</u> PMID: 27091991
- Kohanski MA, Dwyer DJ, Hayete B, Lawrence CA, Collins JJ. A common mechanism of cellular death induced by bactericidal antibiotics. Cell. 2007; 130:797–810. <u>https://doi.org/10.1016/j.cell.2007.06.049</u> PMID: 17803904
- Dwyer DJ, Kohanski MA, Hayete B, Collins JJ. Gyrase inhibitors induce an oxidative damage cellular death pathway in *Escherichia coli*. Mol Syst Biol. 2007; 3:91. <u>https://doi.org/10.1038/msb4100135</u> PMID: 17353933
- Dwyer DJ, Belenky PA, Yang JH, MacDonald IC, Martell JD, Takahashi N, et al. Antibiotics induce redox-related physiological alterations as part of their lethality. Proc Natl Acad Sci USA. 2014; 111: E2100E2109. <u>https://doi.org/10.1073/pnas.1401876111</u> PMID: <u>24803433</u>
- Liu Y, Imlay JA. Cell death from antibiotics without the involvement of reactive oxygen species. Science. 2013; 339:1210–1213. https://doi.org/10.1126/science.1232751 PMID: 23471409
- Maki H, Sekiguchi M. MutT protein specifically hydrolyses a potent mutagenic substrate for DNA synthesis. Nature. 1992; 355:273–275. https://doi.org/10.1038/355273a0 PMID: 1309939
- Zheng Q. rSalvador: An R package for the fluctuation experiment. G3. 2017; 7:3849–3856. <u>https://doi.org/10.1534/g3.117.300120 PMID: 29084818</u>
- Hall BM, Ma CX, Liang P, Singh KK. Fluctuation AnaLysis CalculatOR: a web tool for the determination of mutation rate using Luria-Delbrück fluctuation analysis. Bioinformatics. 2009; 25:1564–1565. <u>https:// doi.org/10.1093/bioinformatics/btp253</u> PMID: <u>19369502</u>
- Gillet-Markowska A, Louvel G, Fischer G. bz-rates: A web tool to estimate mutation rates from fluctuation analysis. G3. 2015; 5:2323–2327. https://doi.org/10.1534/g3.115.019836 PMID: 26338660
- 41. Davis BD. Mechanism of bactericidal action of aminoglycosides. Microbiol Rev. 1987; 51:341. PMID: 3312985
- Drlica K, Malik M, Kerns RJ, Zhao X. Quinolone-mediated bacterial death. Antimicrob Agents Chemother. 2008; 52:385392. https://doi.org/10.1128/AAC.01617-06 PMID: 17724149
- Imlay J, Chin S, Linn S. Toxic DNA damage by hydrogen peroxide through the Fenton reaction in vivo and in vitro. Science. 1988; 240:640642. https://doi.org/10.1126/science.2834821 PMID: 2834821
- 44. Imlay JA. Pathways of oxidative damage. Annu Rev Microbiol. 2003; 57:395418. <u>https://doi.org/10.1146/annurev.micro.57.030502.090938</u> PMID: 14527285
- Couce A, Blázquez J. Side effects of antibiotics on genetic variability. FEMS Microbiol Rev. 2009; 33:531–538. <u>https://doi.org/10.1111/j.1574-6976.2009.00165.x</u> PMID: <u>19260970</u>
- 46. Blázquez J, Couce A, Rodríguez-Beltrán J, Rodríguez-Rojas A. Antimicrobials as promoters of genetic variation. Curr Opin Microbiol. 2012; 15:561–569. <u>https://doi.org/10.1016/j.mib.2012.07.007</u> PMID: <u>22890188</u>
- Treffers HP, Spinelli V, Belser NO. A factor (or mutator gene) influencing mutation rates in *Escherichia coli*. Proc Natl Acad Sci USA. 1954; 40:1064–1071. <u>https://doi.org/10.1073/pnas.40.11.1064</u> PMID: <u>16578437</u>

- Mao EF, Lane L, Lee J, Miller JH. Proliferation of mutators in a cell population. J Bacteriol. 1997; 179:417–422. https://doi.org/10.1128/jb.179.2.417-422.1997 PMID: 8990293
- Sniegowski PD, Gerrish PJ, Lenski RE. Evolution of high mutation rates in experimental populations of E. coli. Nature. 1997; 387:703–705. https://doi.org/10.1038/42701 PMID: 9192894
- Andersson DI, Hughes D. Microbiological effects of sublethal levels of antibiotics. Nat Rev Microbiol. 2014; 12:465–478. https://doi.org/10.1038/nrmicro3270 PMID: 24861036
- Gullberg E, Cao S, Berg OG, Ilbäck C, Sandegren L, Hughes D, et al. Selection of resistant bacteria at very low antibiotic concentrations. PLoS Pathog. 2011; 7:e1002158. <u>https://doi.org/10.1371/journal.ppat.1002158</u> PMID: <u>21811410</u>
- Day T, Read AF. Does high-dose antimicrobial chemotherapy prevent the evolution of resistance? PLoS Comput Biol. 2016; 12: e1004689. <u>https://doi.org/10.1371/journal.pcbi.1004689</u> PMID: 26820986
- Read AF, Day T, Huijben S. The evolution of drug resistance and the curious orthodoxy of aggressive chemotherapy. Proc Natl Acad Sci USA. 2011; 108:10871–10877. <u>https://doi.org/10.1073/pnas.</u> <u>1100299108</u> PMID: <u>21690376</u>
- Day T, Huijben S, Read AF. Is selection relevant in the evolutionary emergence of drug resistance? Trends Microbiol. 2015; 23:126–133. <u>https://doi.org/10.1016/j.tim.2015.01.005</u> PMID: <u>25680587</u>
- 55. Fleming A. Penicillin; 1945. Nobel lecture. Available from: <u>https://www.nobelprize.org/nobel_prizes/</u> medicine/laureates/1945/fleming-lecture.pdf.
- Takano S, Pawlowska BJ, Gudelj I, Yomo T, Tsuru S. Density-dependent recycling promotes the longterm survival of bacterial populations during periods of starvation. mBio. 2017; 8:e02336–16. <u>https://</u> doi.org/10.1128/mBio.02336-16 PMID: 28174316
- Tanouchi Y, Lee AJ, Meredith H, You L. Programmed cell death in bacteria and implications for antibiotic therapy. Trends Microbiol. 2013; 21:265–270. <u>https://doi.org/10.1016/j.tim.2013.04.001</u> PMID: 23684151
- Giroux X, Su WL, Bredeche MF, Matic I. Maladaptive DNA repair is the ultimate contributor to the death of trimethoprim-treated cells under aerobic and anaerobic conditions. Proc Natl Acad Sci USA. 2017; 114:11512–11517. https://doi.org/10.1073/pnas.1706236114 PMID: 29073080
- Myhrvold C, Kotula JW, Hicks WM, Conway NJ, Silver PA. A distributed cell division counter reveals growth dynamics in the gut microbiota. Nat Commun. 2015; 6:10039. <u>https://doi.org/10.1038/</u> ncomms10039 PMID: 26615910
- Price DJ, Breuzé A, Dybowski R, Mastroeni P, Restif O. An efficient moments-based inference method for within-host bacterial infection dynamics. PLoS Comput Biol. 2017; 13: e1005841. <u>https://doi.org/10. 1371/journal.pcbi.1005841</u> PMID: 29155811
- Lyons AB, Parish CR. Determination of lymphocyte division by flow cytometry. J Immunol Methods. 1994; 171:131137. <u>https://doi.org/10.1016/0022-1759(94)90236-4</u> PMID: 8176234
- 62. Callard R, Hodgkin P. Modeling T- and B-cell growth and differentiation. Immunol Rev. 2007; 216:119129. https://doi.org/10.1111/j.1600-065X.2006.00498.x PMID: <u>17367338</u>
- De Boer RJ, Perelson AS. Quantifying T lymphocyte turnover. J Theor Biol. 2013; 327:4587. <u>https://doi.org/10.1016/i.jtbi.2012.12.025</u> PMID: 23313150
- Goss WA, Deitz WH, Cook TM. Mechanism of action of nalidixic acid on *Escherichia coli*. J Bacteriol. 1964; 88:1112–1118. PMID: <u>14219026</u>
- Bos J, Zhang Q, Vyawahare S, Rogers E, Rosenberg SM, Austin RH. Emergence of antibiotic resistance from multinucleated bacterial filaments. Proc Natl Acad Sci USA. 2015; 112:178–183. <u>https://doi.org/10.1073/pnas.1420702111</u> PMID: <u>25492931</u>
- Sun L, Alexander HK, Bogos B, Kiviet DJ, Ackermann M, Bonhoeffer S. Effective polyploidy causes phenotypic delay and influences bacterial evolvability. PLoS Biol. 2018; 16:e2004644. <u>https://doi.org/ 10.1371/journal.pbio.2004644</u> PMID: <u>29470493</u>
- Zheng Q. Toward a unique definition of the mutation rate. Bull Math Biol. 2017; 79:683692. <u>https://doi.org/10.1007/s11538-017-0247-8</u> PMID: <u>28213725</u>
- Gangloff S, Achaz G, Francesconi S, Villain A, Miled S, Denis C, et al. Quiescence unveils a novel mutational force in fission yeast. eLife. 2017; 6:e27469. <u>https://doi.org/10.7554/eLife.27469</u> PMID: 29252184
- Alexander HK, Mayer SI, Bonhoeffer S. Population heterogeneity in mutation rate increases the frequency of higher-order mutants and reduces long-term mutational load. Mol Biol Evol. 2016; 34:419– 436. <u>https://doi.org/10.1093/molbev/msw244</u> PMID: <u>27836985</u>
- Bronner CE, Baker SM, Morrison PT, Warren G, Smith LG, Lescoe MK, et al. Mutation in the DNA mismatch repair gene homologue *hMLH 1* is associated with hereditary non-polyposis colon cancer. Nature. 1994; 368:258–261. <u>https://doi.org/10.1038/368258a0</u> PMID: <u>8145827</u>

- 71. Unna PG. Die Histopathologie der Hautkrankheiten. A. Hirschwald; 1894.
- 72. Hoffman FL. Cancer and smoking habits. Ann Surg. 1931; 93:50–67. PMID: 17866497
- 73. Deichmann U. Germany's forgotten war. Nature. 1999; 401:425–426. https://doi.org/10.1038/46674
- 74. Tomasetti C, Vogelstein B. Variation in cancer risk among tissues can be explained by the number of stem cell divisions. Science. 2015; 347:78–81. <u>https://doi.org/10.1126/science.1260825</u> PMID: 25554788
- Luria SE, Delbrück M. Mutations of bacteria from virus sensitivity to virus resistance. Genetics. 1943; 28:491–511. PMID: <u>17247100</u>
- **76.** Gil D, Bouché JP. ColE1-type vectors with fully repressible replication. Gene. 1991; 105:17–22. PMID: 1937005
- 77. Diard M, Garcia V, Maier L, Remus-Emsermann MNP, Regoes RR, Ackermann M, et al. Stabilization of cooperative virulence by the expression of an avirulent phenotype. Nature. 2013; 494:353–356. <u>https://doi.org/10.1038/nature11913</u> PMID: 23426324
- Kaiser P, Regoes RR, Dolowschiak T, Wotzka SY, Lengefeld J, Slack E, et al. Cecum lymph node dendritic cells harbor slow-growing bacteria phenotypically tolerant to antibiotic treatment. PLoS Biol. 2014; 12: e1001793. <u>https://doi.org/10.1371/journal.pbio.1001793</u> PMID: <u>24558351</u>
- Foster PL. Methods for determining spontaneous mutation rates. Methods Enzymol. 2006; 409:195– 213. <u>https://doi.org/10.1016/S0076-6879(05)09012-9</u> PMID: <u>16793403</u>
- Lea DE, Coulson CA. The distribution of the numbers of mutants in bacterial populations. J Genet. 1949; 49:264–285. <u>https://doi.org/10.1007/BF02986080</u> PMID: <u>24536673</u>
- Zheng Q. A new practical guide to the Luria-Delbrück protocol. Mutat Res. 2015; 781:7–13. <u>https://doi.org/10.1016/j.mrfmmm.2015.08.005</u> PMID: <u>26366669</u>
- 82. Ycart B. Fluctuation analysis with cell deaths. J Appl Probab Stat. 2014; 9:13–29.